Author's response to reviews

Title: Factitious cheilitis: Case report of a rare entity

Authors:

Erdinc Aydin (erdinca@baskent-ank.edu.tr)
Ozgur Gokoglu (ozgurg@baskent-ank.edu.tr)
Gamze Ozcurumez (gamzeo@baskent-ank.edu.tr)
Hakan Aydin (hakana@baskent-ank.edu.tr)

Version: 2 Date: 8 June 2007

Author's response to reviews:

Title page has been added to the manuscript.

"Competing Interests:
The authors declare that they have no competing interests.

Authors' contributions:
OG, GO, HA have been involved in drafting the manuscript. HA involved in pathological, GO in the psychiatric interpretation of the manuscript. EA have involved in revising the manuscript and given the final approval of the version to be published.

Acknowledgements:
Written consent was obtained from the patient for publication of the study," has been added to the manuscript as the editorial office requested.